Author’s response to reviews

Title: Anti-Inflammatory Effects of Olanexidine Gluconate on Oral Epithelial Cells

Authors:

Takuya Nii (nii.takuya@otsuka.jp)
Hiromichi Yumoto (yumoto@tokushima-u.ac.jp)
Katsuhiko Hirota (khirota@kochi-gc.ac.jp)
Yoichiro Miyake (miyake@tks.bunri-u.ac.jp)

Version: 2 Date: 10 Jun 2019

Author’s response to reviews:

We revised our paper according to the reviewers’ reports and corrected the manuscript. We have attached a revised version of the manuscript and our point-by-point responses in a separate document and supplementary information. We believe that the reviewer’s comments allowed us to improve the paper and would like you to convey our gratitude to them.